Author's response to reviews

Title: Prevalence of cutaneous adverse events associated with long-term disease-modifying therapy and their impact on health-related quality of life in patients with multiple sclerosis: a cross-sectional study

Authors:

Deepak M.W. Balak (d.balak@erasusmc.nl)
Gerald J.D. Hengstman (gerald.hengstman@catharinaziekenhuis.nl)
Enes Hajdarbegovic (e.hajdarbegovic@erasusmc.nl)
Rob J.P. van den Brule (rob.vandenbrule@biogenidec.com)
Raymond M.M. Hupperts (r.hupperts@orbisconcern.nl)
Hok Bing Thio (h.thio@erasusmc.nl)

Version: 2 Date: 1 October 2013

Author's response to reviews: see over
Dear Dr Shipley,

Please find enclosed our revised manuscript MS:1727840186105380 entitled “Prevalence of cutaneous adverse events associated with long-term disease-modifying therapy and their impact on health-related quality of life in patients with multiple sclerosis: a cross-sectional study”.

We thank the reviewers and the editors for their time and effort in reviewing our manuscript. Our manuscript has been revised according to the comments raised. The parts that were changed are underlined in the manuscript. Please find below a point-by-point description of the changes made.

Referee 1:
- No changes required.

Referee 2:
- No changes required.

Editorial requirement:
- Please include the names of ethics committees in the manuscript:
  
We have added this information in the methods section at page 4, line 20-22.

On behalf of all the authors I state that this manuscript has not been published and is not being considered for publication elsewhere

Yours sincerely,

Deepak Balak, MD MSc